Author’s response to reviews

Title: How to customize a bona fide psychotherapy for generalized anxiety disorder? A two-arms, patient blinded, ABAB crossed-therapist randomized clinical implementation trial design [IMPLEMENT 2.0]

Authors:
Christoph Flückiger (christoph.flueckiger@psychologie.uzh.ch)
Christine Wolfer (christine.wolfer@uzh.ch)
Judith Held (judith.held@uzh.ch)
Peter Hilpert (hilpert@uw.edu)
Julian Rubel (rubel@uni-trier.de)
Mathias Allemand (m.allemand@uzh.ch)
Richard Zinbarg (rzinbarg@northwestern.edu)
Andreea Vîslă (andreea.visla@uzh.ch)

Version: 2 Date: 02 Mar 2018

Author’s response to reviews:

Dear Sir or Madam

Enclosed please find the revised version of the above-mentioned manuscript.

--> We now deleted footnote 1 and included the text into the manuscript.

Thank you for your contribution during the submission process!

Best, Christoph Flückiger